Dear Editor,

We are honestly grateful for your reviewer suggestions concerning our paper untitled: "TISSUE DETECTION OF NATURAL KILLER CELLS IN COLORECTAL ADENOCARCINOMA"

Please find below a point by point response to the reviewer comments.

In the meantime we did our best to comply accordingly.

Reply to Dr Lachter
1. We made extensive changes in the English language throughout the text.
2. We present our data in a much shortened form (mainly by shortening the Discussion) and the changes are underlined into the text.

Thanking you in advance

Sincerely

Dr. Papanikolaou - Dr. Apostolopoulos